Author's response to reviews

Title: Cannabis in Medicine: A National Educational Needs Assessment among Canadian physicians

Authors:

Daniel Ziemianski (daniel.ziemianski@muhc.mcgill.ca)
Rielle Capler (rielle@telus.net)
Rory Tekanoff (rory@urbancarehealth.com)
Anais Lacasse (Anais.Lacasse@uqat.ca)
Francesca Luconi (francesca.luconi@mcgill.ca)
Mark Ware (mark.ware@mcgill.ca)

Version: 2 Date: 1 July 2014

Author's response to reviews:

To whom it may concern

This revision is submitted at the request of the editors with only the line and page numbers added. No additional changes have been made.

Regards

Dr Mark Ware